ACE Weekly Report : 9/01/99








Spacecraft Activities:


- All spacecraft subsystems are performing nominally.








Maneuver Activities


8/30/99 - An attitude maneuver was successfully executed


Maneuver Start Time=16:02:05     Maneuver End Time=16:09:03


Pre-Maneuver    HGA = -3.72 deg.     Sun Angle = +10.33


Post-Maneuver  HGA = +4.42     Sun Angle =  +5.33








 Instrument Activities 


- 8/30/99 - SWEPAM weekly calibrations performed successfully.








Anomalies


-8/25/99 - The Oracle disk drive in the MOR failed and had to be replaced by H/W maintenance





- 8/26/99 - FOT was unable to bring up DPS after previous day's replacement of Oracle disk drive.  Problem was traced to a corrupted copy of ACE database.  DPS was not ready at beginning of pass, so FOT moved operations to the MAR string 10 minutes before BOT.  Acquired satellite in MOR but did all commanding and all of recorder dump from MAR.  String handover went smoothly.  No commanding or spacecraft problems.  DPS fixed and FOT returned real-time operations to the MOR after pass.  The database files were re-instated using the database back-up tape that the FOT performs periodically





- 8/27/99 - Station had to switch to B/U Signal Processing Center (SPC) equipment. Non-impacting no DR requested.





- 8/28/99 - Intermittent telemetry reception from approx. 1632Z until end of pass.  DR#G05255





- 8/29/99 - Two H/W problems at station 16 - the first was with the prime MFR, the station had to decrease the MFR loop bandwith to improve signal processing quality.  The second problem was with the 26 meter x-axis haulting for 6 seconds. Prime MFR DR number G05258 - X-axis DR number G05259, all data was recovered during realtime support.  AR number G99-0050





- 8/30/99 - Telemetry was interupted from station for ~15 minutes due to a SCP failure at the station.  See DR# G05262 and Anomaly report #G99051





- 8/31/99 - Station 24
